

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2024GL111047

Key Points:

- Meltwater due to Antarctic ice-mass loss results in sea-surface cooling to the southeast of New Zealand in historical and future scenarios
- Including additional Antarctic meltwater in climate model simulations increases wintertime surface westerly winds south of New Zealand
- The magnitude of the response is uncertain due to the wide spread in estimates of observed and projected Antarctic mass imbalance

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. G. Pauling, andrew.pauling@otago.ac.nz

Citation:

Pauling, A. G., Smith, I. J., Ridley, J. K., Martin, T., Thomas, M., & Stevens, D. P. (2025). Impacts of Antarctic ice mass loss on New Zealand climate. *Geophysical Research Letters*, *52*, e2024GL111047. https://doi.org/10.1029/2024GL111047

Received 3 JUL 2024 Accepted 24 JAN 2025

© 2025. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Impacts of Antarctic Ice Mass Loss on New Zealand Climate

Andrew G. Pauling¹, Inga J. Smith¹, Jeff K. Ridley², Torge Martin³, Max Thomas¹, and David P. Stevens^{4,5}

¹Department of Physics, University of Otago, Dunedin, New Zealand, ²Hadley Centre, UK Met Office, Exeter, UK, ³GEOMAR Helmholtz Centre for Ocean Research Kiel, Kiel, Germany, ⁴Centre for Ocean and Atmospheric Sciences, University of East Anglia, Norwich, UK, ⁵School of Mathematics, University of East Anglia, Norwich, UK

Abstract We investigate the impacts of meltwater from Antarctic Ice Sheet (AIS) mass loss on New Zealand climate in a state-of-the-art global climate model. We conduct simulations with additional meltwater from AIS mass loss for both the historical period and a high-emissions future scenario. The ocean surface to the southeast of New Zealand cools, with the largest change in winter and spring. The additional meltwater results in a northward shift of the oceanic sub-tropical front near New Zealand, which partially offsets the projected southward shift of this front in a warming climate. Wintertime surface westerly winds to the south of New Zealand also increase with the addition of the meltwater. The magnitude of the impact of Antarctic meltwater is uncertain due to the wide spread in estimates of Antarctic mass imbalance, but has important implications for future projections for New Zealand climate.

Plain Language Summary Meltwater entering the Southern Ocean due to the Antarctic ice sheet losing mass is not typically included in climate model simulations. We investigate the effect of this missing meltwater on New Zealand climate with a state-of-the-art climate model. We ran simulations with additional meltwater for both the recent past and a projection of the future assuming high greenhouse gas emissions. While the impacts over the land of New Zealand are small, the meltwater causes cooling to the southeast of New Zealand, which offsets the projected warming in this region and has important implications for the local ocean. Wintertime westerly winds to the south of New Zealand also increase. While the exact amount of meltwater entering the Southern Ocean from Antarctica in the recent past and into the future is uncertain, we have shown it may have important impacts on New Zealand climate.

1. Introduction

Mass loss from the Antarctic continent has increased over the last few decades (Slater et al., 2021), and is expected to increase further in the future (Seroussi et al., 2020). This mass loss represents a large source of fresh meltwater to the Southern Ocean, and has been shown to drive surface cooling in the Southern Ocean and Antarctic sea ice expansion in climate model experiments that include it (e.g., Bintanja et al., 2013, 2015; Bronselaer et al., 2018; Mackie, Smith, Ridley, et al., 2020; Mackie, Smith, Stevens, et al., 2020; Pauling et al., 2016, 2017; Swart & Fyfe, 2013; Thomas et al., 2023). There has been disagreement between different modeling studies about the magnitude of the effect of this meltwater on Antarctic sea ice and the Southern Ocean, and determining the reasons for these discrepancies has been hindered by different studies using different amounts, spatial distributions and temporal distributions of freshwater input. This has led to recent efforts to conduct a model intercomparison study with various standardized freshwater input scenarios, the Southern Ocean Freshwater Input from Antarctic Initiative (SOFIA, Swart et al., 2023).

Schmidt et al. (2023) conducted simulations in the GISS-E2.1-G coupled climate model with added meltwater around Antarctica and Greenland as estimated from observational data by Mankoff et al. (2021) and Slater et al. (2021). They found that including the meltwater brought the modeled Antarctic sea ice and Southern Ocean sea surface temperature (SST) trends closer to observed trends than in the simulations without the meltwater. They argue that this should motivate the inclusion of meltwater from ice sheet mass imbalance as a standard historical forcing for all models in the next phase of the Coupled Model Intercomparison Project (CMIP).

The Southern Ocean cooling induced by Antarctic ice sheet mass loss has been shown to have far-reaching impacts, for example, through teleconnections to the tropical eastern Pacific (Dong et al., 2022). New Zealand and its exclusive economic zone (EEZ) sit at the interface between subtropical and subantarctic waters (delineated by the sub-tropical front), which are projected to undergo different changes in a warmer future (Bopp et al., 2013;

Law et al., 2018; Rickard et al., 2016). In the subtropical waters to the north primary productivity is projected to decrease, while productivity may increase in the subantarctic waters to the south. Global warming (without enhanced melting) is projected to shift the sub-tropical front poleward (e.g., Yang et al., 2020), while the Southern Ocean cooling and freshening due to enhanced freshwater input from Antarctica is expected to shift this ocean front equatorward. Due to its proximity to the Southern Ocean and Antarctica, the climate of New Zealand is likely to be strongly influenced by changes in these regions, and so we ask: how does Antarctic ice sheet mass loss impact the climate of New Zealand?

Projections with climate models that participated in the Coupled Model Intercomparison Project Phase 5 (CMIP5, Taylor et al., 2012) show that New Zealand can expect sea surface temperatures to increase by up to 1.1°C by 2100 under a medium-emissions scenario, and up to 2.5°C under a high-emissions scenario (Law et al., 2018). Near-surface air temperatures are projected to increase by 1.4°C(3.0°C) under a medium(high)-emissions scenario (Ministry for the Environment, 2018). The frequency of "hot days" (maximum temperature above 25°C) is projected to increase by up to a factor of 4 under a high-emissions scenario, and droughts and extreme wind events are projected to increase over much of the country. Gibson et al. (2024) found that models participating in CMIP6 (Eyring et al., 2016) project an increase in wintertime precipitation for the west coast of the South Island of New Zealand, with inconsistent changes in the North Island and in summer. Neither CMIP5 nor CMIP6 simulations included meltwater input and the associated Southern Ocean cooling due to projected Antarctic ice-sheet mass loss. In this study we aim to quantify the effect of this missing meltwater on the historical and future climate of New Zealand.

2. Data and Methods

The model experiments used in this study follow the forcing protocol defined in the *hist-antwater-92-11* and *ssp585-ismip6-water* "Tier 2" experiments defined as part of the Southern Ocean Freshwater Input from Antarctica Initiative (SOFIA, Swart et al., 2023). In *hist-antwater-92-11* the additional freshwater input ramps linearly from 0 Sv in 1992 at a rate of 1.1×10^{-3} Sv yr⁻¹ until 2020, at which point the input rate is 0.022 Sv. In the *ssp585-ismip6-water* experiment the freshwater input is prescribed over the period 2015–2100 according to Antarctic mass loss rates from the Ice Sheet Model Intercomparison Project Phase 6 (ISMIP6, Seroussi et al., 2020), reaching approximately 0.2 Sv by the end of the 21st century. These experiments were chosen due to them having the largest meltwater forcing, thus giving an upper bound (based on the SOFIA protocol) on the effect of meltwater on New Zealand climate, and maximizing the signal-to-noise-ratio in the response.

The model used in this study is the CMIP6 model HadGEM3-GC3.1-LL (Kuhlbrodt et al., 2018), which forms the physical core of the New Zealand Earth System Model (Behrens et al., 2020). The NEMO ocean (Gurvan et al., 2019) and GSI8.1 sea ice model (Ridley et al., 2018) components were run at ORCA1 (nominally 1°) horizontal resolution, with 75 vertical layers in the ocean. The Unified Model (UM) atmosphere model component (Walters et al., 2019) was run at $1.875^{\circ} \times 1.25^{\circ}$ horizontal resolution, with 85 vertical layers. This model version reproduces the observed global climate similarly well to the medium-resolution version of HadGEM3-GC3.1 (Williams et al., 2018), while having an order of magnitude lower computational cost. It has also been shown to reproduce Southern Ocean properties well relative to the medium-resolution version of the same model (Kuhlbrodt et al., 2018) and relative to other models with a 1° ocean resolution (Thomas et al., 2023). Importantly for New Zealand climate, the Southern Ocean warm bias, a common problem among coupled climate models, is reduced in the low-resolution version of the model (Kuhlbrodt et al., 2018).

We compare our simulations to the historical and SSP5-8.5 experiments from the Coupled Model Intercomparison Project Phase 6 (CMIP6, Eyring et al., 2016; O'Neill et al., 2016). We also make use of the SSP2-4.5 experiment to extend the historical simulations from the end of 2014 to the end of 2020, as specified in the SOFIA protocol. We compare the model output to the ERA5 reanalysis (Hersbach et al., 2020).

The simulations for this work were run using the identical forcing and model setup as in the CMIP6 historical, ssp245 and ssp585 simulations for HadGEM3-GC3.1-LL, except for the addition of the meltwater from Antarctica as described in the SOFIA protocol. The ssp245 simulations were used to extend the historical simulations from their CMIP6 end date in 2014 until the end of 2020. We ran 10 ensemble members for the *hist-antwater-92-11* experiment, corresponding to members r1i1p1f3–r5i1p1f3 and r11i1p1f3–r15i1p1f3 from the HadGEM3-GC31-LL CMIP6 historical and ssp2-4.5 simulations. We ran 4 ensemble members for the *ssp585*-



Figure 1. Trend in (a) 2 m air temperature, (b) sea surface temperature, and (c) and precipitation over the period 1979–2020 in the ERA5 reanalysis. Stippling denotes where the trend is statistically significant at the 95% confidence level.

ismip6-water experiment, corresponding to ensemble members r1i1p1f3-r4i1p1f3 from the HadGEM-GC31-LL CMIP6 ssp585 simulations. For *hist-antwater-92-11* and *ssp585-ismip6-water* results are presented as the difference between the experiment and its corresponding CMIP6 run with no additional meltwater averaged over the all ensemble members for the last 10 years (2011–2020) or 30 years (2071–2100) of the simulations respectively.

The region of interest for New Zealand climate used in this study is 145°E to 160°W, 60°S to 20°S, which encompasses the New Zealand Exclusive Economic Zone (EEZ), as defined in Law et al. (2018). Statistical significance of the anomalies shown in this paper was computed using a 2-tailed Student's *t*-test.

3. Results

We first examine how climate has changed in New Zealand over the satellite era. The ERA5 reanalysis shows that 2 m air temperatures and SSTs have warmed around, over, and to the north of New Zealand, with no statistically significant trends to the south (Figures 1a and 1b). Trends in precipitation are not statistically significant over or near New Zealand, with some areas of significant increase well to the north (Figure 1c).

We next analyze how well HadGEM3-GC3.1-LL reproduces the observed 2 m air temperature, sea surface temperature (SST), precipitation, and surface wind speed evolution over the satellite era. While the ensemblemean trends in the model may not necessarily resemble the observed trends (Figure 2), the ensemble members with the highest correlation to the observed patterns do capture the observed trends (Figures 2b, 2e, 2h, and 2k). This is not surprising since observations represent a single realization of internal variability, and so we do not necessarily expect the average over many realizations to exhibit a similar spatial pattern. The magnitude of the warming is slightly larger in the model than in observations, likely due to HadGEM3-GC31-LL having a high effective climate sensitivity (Andrews et al., 2019).

The addition of meltwater due to Antarctic ice-mass loss results in cooling of both 2 m air temperatures and SSTs to the southeast of New Zealand in both the *hist-antwater-92-11* and *ssp585-ismip6-water experiments* (Figures 3a–3d). This cooling is strongest in winter and spring in both the historical and future scenarios, with little significant response in the summer (see Figures S1 and S2 in Supporting Information S1). In the *hist-antwater-92-11* experiment the sea surface around Tasmania and extending into the Tasman Sea warms, possibly indicating a change in the East Australian Current. Analysis of the trend in sea surface temperature shows that the addition of meltwater changes the trend such that the model trend become more consistent with the trend in the ERA5 reanalysis (see Figures S5 and S6 in Supporting Information S1). The cooling to the southeast of New Zealand in the *ssp585-ismip6-water* experiment is consistent with a northward shift of the current in that region, induced by the cooling of the Southern Ocean farther south in response to the meltwater.

To investigate the cause of the temperature changes around New Zealand in both the *hist-antwater-92-11* and *ssp585-ismip6-water* experiments, we examine the sea surface height (SSH) and ocean surface velocity responses (Figures 3e, 3f, 3g, and 3h). SSH is an integral quantity linking surface temperature to changes in ocean heat content and currents (divergence of vertically integrated velocity). We see an increase in SSH near Tasmania that extends to the east, mirroring the temperature response in that region in the *hist-antwater-92-11* experiment. There is also an increase in the strength of the East Australian Current that is advecting warm waters to the south. The strengthening of this current is also consistent with an increase in the wind stress curl in the Tasman Sea (not shown). This indicates that the warming in the Tasman Sea is circulation-driven, and is due to changes in the East Australian Current (Figure 3b). This change in circulation and warming is not present in the *ssp585-ismip6-water*



Geophysical Research Letters



Figure 2. Trend in 2 m air temperature, sea surface temperature, precipitation, and wind speed over the period 1979–2020 for (a, d, g, j) the HadGEM3-GC31-LL historical simulation ensemble mean, (b, e, h, k) the HadGEM3-GC31-LL ensemble member with the highest correlation to the ERA5 trend (c, f, i, l) ERA5 reanalysis (repeated from Figure 1 for ease of comparison). Stippling denotes where the trend is statistically significant at the 95% confidence level.

response, indicating that it may be a transient feature of the response that is not present after a few decades. The cooling to the east of New Zealand, extending southeast from the Cook Strait, is also consistent with the decrease in SSH and circulation changes in that region. There is flow from the south into the current flowing east from Cook Strait that brings cooler waters to this region. The cooling to the South of New Zealand shifts the SSH gradient north, resulting in cooling in the region of the current emanating from Cook Strait.

Antarctic meltwater does not have a strong impact on average (see Figure S4 in Supporting Information S1) or extreme (not shown) precipitation for New Zealand. This is perhaps not surprising, given the highly variable nature of precipitation around New Zealand and its lack of response to anthropogenic climate change seen in the reanalysis (Figure 1c). In the CMIP6 ssp585 simulations with no additional Antarctic meltwater the average number of hot days over land (daily maximum 2 m air temperature ≥ 25 °C) increases by about 27 days per year for the period 2081–2100 relative to 2001–2020. The addition of the Antarctic meltwater results in a statistically significant reduction of the projected change by 2.1 days per year on average, with the largest changes in the North Island (see Figure S5 in Supporting Information S1).

Projected changes to near-surface winds are also expected to be an important consequence of a warming climate for New Zealand. Thus, we examine the projected changes in surface wind speed with the addition of Antarctic meltwater. While there is little response in seasonal-mean wind speed over the land of New Zealand, there is a significant increase in wind speed to the south of New Zealand in winter in the *hist-antwater-92-11* experiment (Figure 4c). This change is dominated by an increase in the westerly component of the winds (not shown). This wintertime wind speed increase is also found, albeit weaker and farther south, in the *ssp585-ismip6-water*. This is

4 of 9





Figure 3. (a, b) 2 m air temperature, (c, d) sea surface temperature, (e, f) sea surface height and (g, h) ocean surface velocity response for the *hist-antwater-92-11* and *ssp585-ismip6-water* experiments. The response is computed as the difference between each experiment and its CMIP6 control run with no additional meltwater averaged over 2011–2020 for *hist-antwater-92-11* or 2071–2100 for *ssp585-ismip6-water*. Stippling denotes where the response is statistically significant at the 95% confidence level.



Geophysical Research Letters

10.1029/2024GL111047



Figure 4. Seasonal-mean near-surface wind speed response in *hist-antwater-92-11* (a–d) and *ssp585-ismip6-water* (e–h). Stippling denotes where the anomaly is statistically significant at the 95% confidence level.

consistent with the poleward shift of the westerly winds that is expected as the climate warms (e.g., Perren et al., 2020). The wind-speed increase in *hist-antwater-92-11* is consistent with the increased local meridional SST gradient seen in Figure 3c, which tends to lead to stronger zonal winds due to increased baroclinity (e.g., Hoskins & Hodges, 2005). This increase in local SST gradient is not present in the future scenario, hence we do not see the same change in winds. We also examined wind extremes, by computing the change in the 99th percentile daily-mean wind speed. We found no significant difference in the change in wind extremes by the end of the 21st century with the addition of Antarctic meltwater (not shown).

4. Discussion and Conclusions

The cooling to the southeast of New Zealand in response to Antarctic meltwater (Figure 3) is consistent with the response seen in Schmidt et al. (2023). This provides some evidence that this response is robust to the choice of climate model and reinforces the call of Schmidt et al. (2023) for meltwater due to Antarctic ice mass loss to be included as a historical forcing in CMIP7. Omitting this meltwater from historical simulations and future projections will bias temperatures and winds to the south of New Zealand. The studies of Rickard et al. (2016) and Law et al. (2018) highlight the complex nature of the relationship between climate and ocean biogeochemistry in this region, thus model simulations with both added Antarctic meltwater and an active biogechemistry model component are needed to understand the potential impacts on primary productivity.

The model used in this study, HadGEM3-GC3.1-LL, has been found to exhibit spurious open-ocean deep convection in the Southern Ocean (Ridley et al., 2022). The addition of Antarctic meltwater to the Southern Ocean has been shown to substantially reduce this deep convection in models (Chen et al., 2023). To ensure that our results are not influenced by this change in open-ocean deep convection through teleconnections to the New Zealand region, we computed our results with the two historical ensemble members exhibiting substantial Southern Ocean deep convection (out of 10 total ensemble members) removed. This resulted in little change to the response (not shown), indicating that the results presented here are not strongly influenced by spurious deep convection in the model.

There is high uncertainty in the magnitude of the meltwater input used in both the historical and future scenarios presented here. The historical meltwater forcing is based on the observational estimates of Slater et al. (2021). These estimates are sufficiently uncertain that it motivated the SOFIA team to create four different historical meltwater scenarios with varying start dates and rates of change based on the same underlying data. We chose the scenario with the highest rate of increase in meltwater input, in order to provide an upper bound on the response and to maximize the signal-to-noise ratio. We expect that choosing one of the other scenarios would result in smaller impacts in the New Zealand region. The meltwater input for the future scenarios was based on the model output from the Ice Sheet Model Intercomparison for CMIP6 (ISMIP6, Seroussi et al., 2020; Thomas et al., 2023). As can be seen in Figure A1 from Swart et al. (2023), there is a large inter-model spread in meltwater input from



Geophysical Research Letters



Figure 5. Contours defining the sub-tropical front (STF) around New Zealand for (a, b) the first 20 years and the last 20 years of the ssp585 simulations. (c, d) The *ssp585-ismip6-water* experiment versus ssp585 averaged over 2081–2100. The left column shows the STF as defined by the 13 °C temperature contour at 100 m depth, and the right column the 34 psu contour at 100 m depth. The shaded region shows the range in STF location given by varying the temperature threshold by ± 1 °C or the salinity threshold by ± 0.2 psu.

the ISMIP6 models, particularly toward the end of the 21st century. The uncertainty in future projections of Antarctic meltwater input can also be seen in the wide range of projections used in previous studies. For example, Bronselaer et al. (2018) used meltwater input from the ice sheet model of DeConto and Pollard (2016), which reaches ~0.6 Sv by the end of the 21st century, about three times that used in this study. While such high estimates of ice sheet mass loss have recently been called into question (e.g., Morlighem et al., 2024), this uncertainty means that the response to Antarctic meltwater could be substantially stronger or weaker than that seen in this study. Continued effort into ice sheet modeling is needed to narrow this uncertainty range. Further uncertainty in future projections of meltwater comes from the choice of emissions scenario. We chose the meltwater input from ice sheet model simulations forced with climate from the SSP5-8.5 scenario, a high-emissions scenario that is likely unrealistic (Hausfather & Peters, 2020). We chose this scenario to provide an upper bound (based on the SOFIA protocol) on the magnitude of the response to Antarctic meltwater and to maximize the signal-to-noise ratio. Choosing a meltwater input from ISMIP6 models forced with a lower-emissions scenario would result in less additional meltwater input to the Southern Ocean and thus a different response for the New Zealand region.

The meltwater-induced cooling to the south of New Zealand may have important implications for the local ocean. New Zealand sits at the boundary between the sub-tropical waters to the north and sub-antarctic waters to the south. The sub-tropical front (STF) delineates these two regions, and the position of this front has important implications for the local ocean conditions around New Zealand, as temperature can vary by $4-5^{\circ}$ C, and salinity by 1 psu across a distance of less than 200 km (Belkin & Gordon, 1996). The STF has varying definitions, but is typically defined by the southernmost region meeting a certain salinity or temperature threshold at ~ 100 m depth (e.g., Behrens et al., 2021; Orsi et al., 1995). Due to both model biases in the mean state temperature and salinity distribution and the substantially warmer ocean in the ssp585 scenario, the temperature and salinity thresholds used here (13°C and 34 psu) differ from those used in Behrens et al. (2021) (11°C and 34.8 psu). These thresholds were chosen to most closely follow the position of greatest temperature or salinity gradient in the model fields. The STF, in particular that defined according to the temperature threshold is expected to move southward as the climate warms (Yang et al., 2020, and Figures 5a and 5b). The inclusion of Antarctic meltwater and its associated Southern Ocean cooling partially offset this shift (Figures 5c and 5d), with the position of the STF slightly farther

19448007, 2025, 4, Downloaded

from https://agupubs.onlinelibrary

north in *ssp585-ismip6-water* than ssp585 whether defined by the temperature or salinity threshold. This further highlights the importance of including Antarctic meltwater in climate projections for the New Zealand region.

This study provides the first quantitative estimate of the effect of including meltwater from Antarctic ice-mass loss for New Zealand climate. There is significant cooling in all seasons to the southeast of New Zealand in both historical and future scenarios, while the response over the land of New Zealand is weak. There is also significant warming in the Tasman Sea in the historical scenario that is not present in the future, suggesting a transient impact on the East Australian Current. While wind changes over the land of New Zealand are small, there is a significant increase in wintertime surface wind speed to the south. Including Antarctic meltwater in model simulations of climate is important for regions close to the Southern Ocean in particular, and we echo recent calls for it to be included as a forcing in future simulations, until such time as realistic ice-sheet models can be coupled to state-of-the-art climate models.

Data Availability Statement

CMIP6 data used in this study is available from the Earth System Grid Federation (ESGF) archive: https://aims2. llnl.gov/search/cmip6. The SOFIA model output data are available at https://crd-data-donnees-rdc.ec.gc.ca/ CCCMA/SOFIA (Swart et al., 2023). Code and processed data necessary to reproduce the results of this study are archived at Pauling (2024a, 2024b) respectively.

References

- Andrews, T., Andrews, M. B., Bodas-Salcedo, A., Jones, G. S., Kuhlbrodt, T., Manners, J., et al. (2019). Forcings, feedbacks, and climate sensitivity in HadGEM3-GC3.1 and UKESM1. *Journal of Advances in Modeling Earth Systems*, 11(12), 4377–4394. https://doi.org/10.1029/ 2019MS001866
- Behrens, E., Hogg, A. M., England, M. H., & Bostock, H. (2021). Seasonal and interannual variability of the subtropical front in the New Zealand region. *Journal of Geophysical Research: Oceans*, 126(2), e2020JC016412. https://doi.org/10.1029/2020JC016412
- Behrens, E., Williams, J., Morgenstern, O., Sutton, P., Rickard, G., & Williams, M. J. M. (2020). Local grid refinement in New Zealand's Earth System Model: Tasman Sea ocean circulation improvements and super-gyre circulation implications. *Journal of Advances in Modeling Earth* Systems, 12(7), e2019MS001996. https://doi.org/10.1029/2019MS001996
- Belkin, I. M., & Gordon, A. L. (1996). Southern Ocean fronts from the Greenwich meridian to Tasmania. Journal of Geophysical Research, 101(C2), 3675–3696. https://doi.org/10.1029/95JC02750
- Bintanja, R., van Oldenborgh, G. J., Drijfhout, S. S., Wouters, B., & Katsman, C. A. (2013). Important role for ocean warming and increased iceshelf melt in Antarctic sea-ice expansion. *Nature Geoscience*, 6(5), 376–379. https://doi.org/10.1038/NGEO1767
- Bintanja, R., Van Oldenborgh, G. J., & Katsman, C. A. (2015). The effect of increased fresh water from Antarctic ice shelves on future trends in Antarctic sea ice. *Annals of Glaciology*, 56(69), 120–126. https://doi.org/10.3189/2015AoG69A001
- Bopp, L., Resplandy, L., Orr, J. C., Doney, S. C., Dunne, J. P., Gehlen, M., et al. (2013). Multiple stressors of ocean ecosystems in the 21st century: Projections with CMIP5 models. *Biogeosciences*, 10(10), 6225–6245. https://doi.org/10.5194/bg-10-6225-2013
- Bronselaer, B., Winton, M., Griffies, S. M., Hurlin, W. J., Rodgers, K. B., Sergienko, O. V., et al. (2018). Change in future climate due to Antarctic meltwater. *Nature*, 564(7734), 53–58. https://doi.org/10.1038/s41586-018-0712-z
- Chen, J.-J., Swart, N. C., Beadling, R., Cheng, X., Hattermann, T., Jüling, A., et al. (2023). Reduced deep convection and bottom water formation due to Antarctic meltwater in a multi-model ensemble. *Geophysical Research Letters*, 50(24), e2023GL106492. https://doi.org/10.1029/ 2023GL106492
- DeConto, R. M., & Pollard, D. (2016). Contribution of Antarctica to past and future sea-level rise. Nature, 531(7596), 591–597. https://doi.org/10. 1038/nature17145
- Dong, Y., Pauling, A. G., Sadai, S., & Armour, K. C. (2022). Antarctic ice-sheet meltwater reduces transient warming and climate sensitivity through the sea-surface temperature pattern effect. *Geophysical Research Letters*, 49(24), e2022GL101249. https://doi.org/10.1029/ 2022GL101249
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Gibson, P. B., Rampal, N., Dean, S. M., & Morgenstern, O. (2024). Storylines for future projections of precipitation over New Zealand in CMIP6 models. *Journal of Geophysical Research: Atmospheres*, 129(5), e2023JD039664. https://doi.org/10.1029/2023JD039664
- Gurvan, M., Bourdallé-Badie, R., Bouttier, P.-A., Bricaud, C., Bruciaferri, D., Calvert, D., et al. (2019). NEMO ocean engine. Zenodo. https://doi.org/10.5281/zenodo.3248739
- Hausfather, Z., & Peters, G. P. (2020). Emissions—The "business as usual" story is misleading. *Nature*, 577(7792), 618–620. https://doi.org/10. 1038/d41586-020-00177-3
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Hoskins, B. J., & Hodges, K. I. (2005). A new perspective on Southern Hemisphere storm tracks. Journal of Climate, 18(20), 4108–4129. https:// doi.org/10.1175/JCL13570.1
- Kuhlbrodt, T., Jones, C. G., Sellar, A., Storkey, D., Blockley, E., Stringer, M., et al. (2018). The low-resolution version of HadGEM3 GC3.1: Development and evaluation for global climate. *Journal of Advances in Modeling Earth Systems*, 10(11), 2865–2888. https://doi.org/10.1029/ 2018MS001370

Acknowledgments

AGP, IJS, and MT were funded by the Deep South National Science Challenge (MBIE contract number C01X1412). AGP, US, and MT acknowledge additional support from the Antarctic Science Platform (University of Otago subcontract 19424 from VUW's ASP Project 4 contract with Antarctica New Zealand through MBIE SSIF Programmes Investment contract number ANTA1801). JR was supported by the Met Office Hadley Centre Climate Programme funded by DBEIS and Defra. The authors wish to acknowledge the use of New Zealand eScience Infrastructure (NeSI) high performance computing facilities and consulting support as part of this research. New Zealand's national facilities are provided by NeSI and funded jointly by NeSI's collaborator institutions and through the Ministry of Business, Innovation & Employment's Research Infrastructure programme. URL: https://www.nesi.org. nz. We acknowledge the World Climate Research Programme, which coordinated and promoted CMIP6. We thank the climate modeling groups for producing and making available their model output, the Earth System Grid Federation (ESGF) for archiving the data and providing access, and the multiple funding agencies who support CMIP6 and ESGF. The authors would like to thank Jonny Williams for technical support while conducting the model simulations in this study. We also thank the wider SOFIA team for useful discussions over the course of this project. Open access publishing facilitated by University of Otago, as part of the Wiley -University of Otago agreement via the Council of Australian University Librarians

- Law, C. S., Rickard, G. J., Mikaloff-Fletcher, S. E., Pinkerton, M. H., Behrens, E., Chiswell, S. M., & Currie, K. (2018). Climate change projections for the surface ocean around New Zealand. New Zealand Journal of Marine and Freshwater Research, 52(3), 309–335. https://doi.org/ 10.1080/00288330.2017.1390772
- Mackie, S., Smith, I. J., Ridley, J. K., Stevens, D. P., & Langhorne, P. J. (2020). Climate response to increasing Antarctic iceberg and ice shelf melt. Journal of Climate, 33(20), 8917–8938. https://doi.org/10.1175/JCLI-D-19-0881.1
- Mackie, S., Smith, I. J., Stevens, D. P., Ridley, J. K., & Langhorne, P. J. (2020). Interactions between increasing CO₂ and Antarctic melt rates. *Journal of Climate*, 33(20), 8939–8956. https://doi.org/10.1175/JCLI-D-19-0882.1
- Mankoff, K. D., Fettweis, X., Langen, P. L., Stendel, M., Kjeldsen, K. K., Karlsson, N. B., et al. (2021). Greenland ice sheet mass balance from 1840 through next week. *Earth System Science Data*, 13(10), 5001–5025. https://doi.org/10.5194/essd-13-5001-2021
- Ministry for the Environment. (2018). Climate change projections for New Zealand: Atmosphere projections based on simulations from the IPCC fifth assessment (2nd ed.).
- Morlighem, M., Goldberg, D., Barnes, J. M., Bassis, J. N., Benn, D. I., Crawford, A. J., et al. (2024). The West Antarctic Ice Sheet may not be vulnerable to marine ice cliff instability during the 21st century. *Science Advances*, 10(34), eado7794. https://doi.org/10.1126/sciadv.ado7794
 O'Neill, B. C., Tebaldi, C., van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The Scenario Model Intercomparison Project
- (ScenarioMIP) for CMIP6. *Geoscientific Model Development*, 9(9), 3461–3482. https://doi.org/10.5194/gmd-9-3461-2016
- Orsi, A. H., Whitworth, T., & Nowlin, W. D. (1995). On the meridional extent and fronts of the Antarctic Circumpolar Current. Deep Sea Research Part I: Oceanographic Research Papers, 42(5), 641–673. https://doi.org/10.1016/0967-0637(95)00021-W
- Pauling, A. G. (2024a). and rewpauling/nzmeltwater: nzmeltwater v2.0 [Software]. Zenodo. https://doi.org/10.5281/zenodo.14159886
- Pauling, A. G. (2024b). Supporting data for "Impacts of Antarctic ice mass loss on New Zealand climate [Dataset]. Zenodo. https://doi.org/10. 5281/zenodo.14110477
- Pauling, A. G., Bitz, C. M., Smith, I. J., & Langhorne, P. J. (2016). The response of the southern ocean and Antarctic sea ice to freshwater from ice shelves in an Earth System Model. *Journal of Climate*, 29(5), 1655–1672. https://doi.org/10.1175/JCLI-D-15-0501.1
- Pauling, A. G., Smith, I. J., Langhorne, P. J., & Bitz, C. M. (2017). Time-dependent freshwater input from ice shelves: Impacts on Antarctic sea ice and the southern ocean in an Earth System Model. *Geophysical Research Letters*, 44(20), 10454–10461. https://doi.org/10.1002/ 2017GL075017
- Perren, B. B., Hodgson, D. A., Roberts, S. J., Sime, L., Van Nieuwenhuyze, W., Verleyen, E., & Vyverman, W. (2020). Southward migration of the Southern Hemisphere westerly winds corresponds with warming climate over centennial timescales. *Communications Earth & Environment*, 1(1), 1–8. https://doi.org/10.1038/s43247-020-00059-6
- Rickard, G. J., Behrens, E., & Chiswell, S. M. (2016). CMIP5 earth system models with biogeochemistry: An assessment for the southwest Pacific Ocean. Journal of Geophysical Research: Oceans, 121(10), 7857–7879. https://doi.org/10.1002/2016JC011736
- Ridley, J. K., Blockley, E. W., & Jones, G. S. (2022). A change in climate state during a pre-industrial simulation of the CMIP6 model HadGEM3 driven by deep ocean drift. *Geophysical Research Letters*, 49(6), e2021GL097171. https://doi.org/10.1029/2021GL097171
- Ridley, J. K., Blockley, E. W., Keen, A. B., Rae, J. G. L., West, A. E., & Schroeder, D. (2018). The sea ice model component of HadGEM3-GC3.1. Geoscientific Model Development, 11(2), 713–723. https://doi.org/10.5194/gmd-11-713-2018
- Schmidt, G. A., Romanou, A., Roach, L. A., Mankoff, K. D., Li, Q., Rye, C. D., et al. (2023). Anomalous meltwater from ice sheets and ice shelves is a historical forcing. *Geophysical Research Letters*, 50(24), e2023GL106530. https://doi.org/10.1029/2023GL106530
- Seroussi, H., Nowicki, S., Payne, A. J., Goelzer, H., Lipscomb, W. H., Abe-Ouchi, A., et al. (2020). ISMIP6 Antarctica: A multi-model ensemble of the Antarctic ice sheet evolution over the 21st century. *The Cryosphere*, 14(9), 3033–3070. https://doi.org/10.5194/tc-14-3033-2020
- Slater, T., Lawrence, I. R., Otosaka, I. N., Shepherd, A., Gourmelen, N., Jakob, L., et al. (2021). Review article: Earth's ice imbalance. *The Cryosphere*, 15(1), 233–246. https://doi.org/10.5194/tc-15-233-2021
- Swart, N. C., & Fyfe, J. C. (2013). The influence of recent Antarctic ice sheet retreat on simulated Sea ice area trends. *Geophysical Research Letters*, 40(16), 4328–4332. https://doi.org/10.1002/grl.50820
- Swart, N. C., Martin, T., Beadling, R., Chen, J.-J., Danek, C., England, M. H., et al. (2023). The Southern Ocean Freshwater Input from Antarctica (SOFIA) Initiative: Scientific objectives and experimental design. *Geoscientific Model Development*, 16(24), 7289–7309. https://doi.org/10. 5194/gmd-16-7289-2023
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Thomas, M., Ridley, J. K., Smith, I. J., Stevens, D. P., Holland, P. R., & Mackie, S. (2023). Future response of Antarctic continental shelf temperatures to ice shelf basal melting and calving. *Geophysical Research Letters*, 50(18), e2022GL102101. https://doi.org/10.1029/ 2022GL102101
- Walters, D., Baran, A. J., Boutle, I., Brooks, M., Earnshaw, P., Edwards, J., et al. (2019). The Met Office Unified Model Global Atmosphere 7.0/ 7.1 and JULES global land 7.0 configurations. *Geoscientific Model Development*, 12(5), 1909–1963. https://doi.org/10.5194/gmd-12-1909-2019
- Williams, K. D., Copsey, D., Blockley, E. W., Bodas-Salcedo, A., Calvert, D., Comer, R., et al. (2018). The Met Office Global Coupled Model 3.0 and 3.1 (GC3.0 and GC3.1) Configurations. *Journal of Advances in Modeling Earth Systems*, 10(2), 357–380. https://doi.org/10.1002/ 2017MS001115
- Yang, H., Lohmann, G., Lu, J., Gowan, E. J., Shi, X., Liu, J., & Wang, Q. (2020). Tropical expansion driven by poleward advancing midlatitude meridional temperature gradients. *Journal of Geophysical Research: Atmospheres*, 125(16), e2020JD033158. https://doi.org/10.1029/ 2020JD033158